Author's response to reviews

Title: Bone marrow-derived cells can acquire renal stem cells properties and ameliorate ischemia-reperfusion induced acute renal injury

Authors:

Xiaohua Jia (yyjtd@163.com)
Xiaoqiang Xie (xxq666@21cn.cm)
Guowei Feng (tianjinfengguowei@126.com)
Qinjun Zhao (zhaoqinjun@hotmail.com)
Yongzhe Che (cheli@nankai.edu.cn)
Zhongchao Han (hanzhongchao@hotmail.com)
Yong Xu (xuyong8816@sina.com)
Zongjin Li (lizongjin@hotmail.com)
Deling Kong (kongdeling@nankai.edu.cn)

Version: 2 Date: 12 January 2012

Author's response to reviews:

Dear Editor,

We are sorry for this. The animal experiments were performed according to the Regulations for the Administration of Affairs Concerning Experimental Animals (Tianjin, June 2004) and Animal protocols were approved by the Nankai University Animal Care and Use Committee. We have clarified this.

zongjin